Author's response to reviews

Title: The need for nursing instruction in patients receiving steroid pulse therapy for the treatment of autoimmune diseases and the effect of instruction on patient knowledge

Authors:

Yu-Chu Pai (ycpai@vghtpe.gov.tw)

Version: 5 Date: 23 August 2010

Author's response to reviews:

Hans Zauner
Scientific Editor, BioMed Central

RE: MS: 6662347482988180
Title: The effect of nursing instruction in patients receiving steroid pulse therapy for the treatment of autoimmune diseases

Dear Mr. Zauner:

It is a please to be writing you. Thank you for your continued review of our manuscript and for the valuable comments of the reviewers. We have clarified the ethical approval we obtained for this study by adding a statement in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

“Before data collection, the purpose of the study was explained to the patients, and the written informed consent was obtained from each patient. This study was conducted in accordance with Good Clinical Practice and the Declaration of Helsinki and its amendments. The protocol was approved by the Institutional Review Board of Taipei Veterans General Hospital (Reference No. 94-03-17A).”

Thank you again for your continued review of our work. We are looking forward to you continued correspondence.

Very sincerely:
Yu-Chu Pai